Author’s response to reviews

Title: Patient access to complex chronic disease records on the Internet

Authors:

Cherry Bartlett (Cherry.Bartlett@kidneycare.nhs.uk)
Keith Simpson (keith.simpson@nhs.net)
A Neil Turner (neil.turner@ed.ac.uk)

Version: 4 Date: 25 July 2012

Author’s response to reviews:

Ref 18 updated to be as contemporary as possible (corresponding numbers amended in 3rd para of Discussion)